Author's response to reviews

Title: Inflammatory cytokine response to exercise in alpha-1-antitrypsin deficient COPD patients 'on' or 'off' augmentation therapy

Authors:

Ivan Mark Olfert (molfert@hsc.wvu.edu)
Moh H Malek (en7488@wayne.edu)
Tomas ML Eagan (tomas.eagan@med.uib.no)
Harrieth Wagner (hewagner@ucsd.edu)
Peter D. Wagner (pdwagner@ucsd.edu)

Version: 2 Date: 2 January 2014

Author's response to reviews:

As requested, we have revised the manuscript to include the following elements,
(1) Title Page with email addresses of all authors.
(2) Conclusion provided after Discussion.
(3) Competing Interests statement.
(4) Authors’ Contributions statement.